Author's response to reviews

Title: Quality of care and 30 day mortality among patients with hip fractures: a nationwide cohort study

Authors:

Katrine A Nielsen (kan@ikas.dk)
Niels C Jensen (nc.jensen@mail1.stofanet.dk)
Claus M Jensen (cmj@dadlnet.dk)
Marianne Thomsen (mth@ribeamt.dk)
Lars Pedersen (lap@dce.au.dk)
Søren P Johnsen (spj@dce.au.dk)
Annette Ingeman (Annette.Ingeman@STAB.RM.DK)
Paul D Bartels (Paul.Bartels@STAB.RM.DK)
Reimar W Thomsen (r.thomsen@rn.dk)

Version: 7 Date: 23 September 2009

Author's response to reviews:

I have checked for typographical errors. For some reason, there are some sentences on page 14 displayed in Times-Roman, rather than in Times New Roman like the rest of the manuscript. I have corrected this and uploaded a revised manuscript main.doc.